Author's response to reviews

Title: Effects of isoflavones (soy phyto-estrogens) on serum lipids: A meta-analysis of randomized controlled trials

Authors:

Dr John Yeung (john@epidemiology.org.hk)
Tak Fu Yu (yutakfu1@yahoo.com.hk)

Version: 3 Date: 9 Nov 2003

Changes were made upon requests on title page.

Also, (95%CI: -1.0 to 2.0, heterogenrity p = 0.5) was changed to (95%CI = -1.0 to 2.0; heterogenrity p = 0.5).